Author's response to reviews

Title: Frequent food insecurity among injection drug users: correlates and concerns

Authors:

Carol J Strike (carol.strike@utoronto.ca)
Katherine Rudzinski (kat.rudzinski@utoronto.ca)
Jessica Patterson (jessica.patterson@utoronto.ca)
Margaret Millson (p.millson@utoronto.ca)

Version: 5 Date: 16 November 2012

Author's response to reviews:

Dear Dr. Heimer,

Thank you sending the reviewer comments and the opportunity to improve the manuscript. Reviewer 1 asked for changes to the formatting of tables 2 and 3. I reformatted table 2. However, as communicated with Mr. Jimmar Dizon from the Journal office, I did not remove the internal lines within the tables for Tables 2 and 3. As the journal guidelines state, these are to be included in tables. Mr. Dizon wrote 'Please ignore the recommendation of Reviewer 1 and follow the journal guidelines'.

Reviewer 2 had no further comments.

I believe we have now addressed all outstanding comments.

Sincerely,

Carol Strike